\def\GW{/GW}
\def\visibility{true}
\input{\GW/root/GW4reports/GWlatexBeamerHead.tex}

\AtBeginSection[]
{
  \begin{frame}
    \frametitle{Table of Contents}
    \tableofcontents[currentsection]
  \end{frame}
}

\AtBeginSubsection[]
{
  \begin{frame}
    \frametitle{Table of Contents}
    \tableofcontents[currentsubsection]
  \end{frame}
}


\begin{document}

\title{Scientific and Education Activities on the GOLEM Tokamak in the Framework of the IAEA CRP}
\author{Vojt\v{e}ch Svoboda \\ on behalf of the tokamak GOLEM team }
\date{Prague, \today}
\slide{\centering \small  2$^{nd}$ Research Coordination Meeting
CRP F13019 
Network of Small and Medium Size Magnetic Confinement Fusion Devices for Fusion Research}{\titlepage}

\section{Introduction}
\GWis{Tokamak/BasicDescription/ParametersWithPhoto}

\slide{Objectives in the Framework of the IAEA CRP}{
\begin{itemize}
\item 2.2.14.1.   Main research  activities proposed:
\begin{itemize}
\item {\bf Plasma edge studies} using advanced probe techniques.
\item Developing {\bf diagnostics for runaway studies}.
\end{itemize} 
\item 2.2.14.3.   Educational   activities:
\begin{itemize}
\item {\bf on-site:} GOMTRAIC, a week of hands-on experiments at the GOLEM tokamak.
\item {\bf remote:} Set of remote participation training courses
\end{itemize} 
\end{itemize} 
(Golem activities ratio: 20\% Research, 80\% Education)
}

\noslide{Objectives: Research}{
\begin{center}
\GWig{width=\tw}{Projects/IAEA/17Network/Events/Research.jpg}
\end{center}
}

\noslide{Objectives: Education}{
\begin{center}
\GWig{width=\tw}{Projects/IAEA/17Network/Events/Education.jpg}
\end{center}
}

\section{Work actually done (selected examples)}
\GWin{Experiments/IzotopicStudies/Slides/GoshaHydrogenHeliumMainResult}\GWslideSetup\GWslideResult

\GWin{Experiments/RunAwayElectronStudies/Diagnostics/Slides/StripDetector}\GWslideSetup\GWslideResult

\GWin{Experiments/BreakDownStudies/Slides/20UkrReport}\GWslideResult


\section{Training events}
\GWin{TrainingCourses/PlasmaSchools/GOMTRAIC.cz/19/Slides}
\Proposal
\LastPhotoWithNumbers
\GWis{Chronicle/Inventory/Remotes/1018-0121}

\slide{New level of remote training: ... with publication on the horizon. }{
\begin{itemize}
\item Workshop "Kiten 2018" : Operational Domain in Hydrogen Plasmas on the GOLEM Tokamak. Journal of Fusion Energy (2019) 38:253–261.
\item Training course for the Kharkiv National University: Breakdown Phase in The Golem Tokamak and  its Impact on Plasma Performance. Ready for publishing in  Ukraine J. Phys. 
\item Training for the National Research Nuclear University "MEPhI", Moscow: Hydrogen and Helium Plasmas in the GOLEM Tokamak. Almost ready for publishing.
\end{itemize}
}

\GWis{TrainingCourses/Production/Slides/PandemicSetup}



\section{Plan for the next period } 

\slide{Plans (adapted for remote regimes) \\ \footnotesize(step forward to the integral experiments)}{
\begin{itemize}
\item 2.2.14.1.   Main research activities proposed:
\begin{itemize}
\item Plasma edge studies using advanced probe techniques.
\begin{itemize}
\item Measurement with double rake electrostatic probe  to find the position of the velocity shear layer (VSL).
\item Operational domains of the GOLEM tokamak.
\item Physics of the plasma edge column with a focus on determining and comparing the position of the plasma edge by electrostatic, magnetic and radiation measurement methods. 
\item Isotopic studies cont. with Sasha at.al.
\item Ion temperature measurement.
\end{itemize}
\item Developing diagnostics for runaway studies.
\begin{itemize}
\item Integral experiments focused on the behavior of runaway electrons, where a wide range of different types of HXR probes will be involved: scintillation, medipix, timepix, calorimetric and strip detectors.
\end{itemize} 
\end{itemize} 
\item 2.2.14.3.   Education activities:
\begin{itemize}
\item GOMTRAIC 2021, a week of hands-on experiments at the GOLEM tokamak (maybe remote version).
\item Set of remote participation training courses (developing pandemic set-up)
\end{itemize} 
\end{itemize} 
}



\nocite{8876584,8876482,8824534,Dhyani_2019,JOFESvoboda2019,DhyaniEPS19,MachaEPS19,GRYAZNEVICH_2020,Novotny_2020}

\begin{frame}[allowframebreaks] \frametitle{Relevant References}
%\bibliographystyle{abbrvnat}
%\bibliographystyle{plain}
\bibliographystyle{apalike}
\creditfont
\bibliography{/GW/root/bibtex/golem,/GW/root/bibtex/others}


\end{frame}

\end{document}


